Author’s response to reviews

Title: Design of a RCT evaluating the (cost-)effectiveness of a lifestyle intervention for male construction workers at risk for cardiovascular disease: The Health under Construction study

Authors:
Iris F Groeneveld (iris.groeneveld@vumc.nl)
Karin I Proper (ki.proper@vumc.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)
Cor van Duivenbooden (vanduivenbooden@arbouw.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)

Version: 3 Date: 2 January 2008

Dear editor,

For the latest version of the manuscript I made the following changes:

Page 1: Iris F Groeneveld1,2, Karin I Proper1,2, Allard J van der Beek1,2, Cor van Duivenbooden3, Willem van Mechelen1,2

Page 3/4: Furthermore, one third of all workers in the construction industry are smokers.

Page 16: ...the 10-point scales about importance, confidence and readiness, the first short-term goals and the long-term goal(s) (step 3, 4 and 5, respectively).

Page 17: ...a 3-day stay in a bungalow park can be won by the participant who best completes a slogan.

Page 18: For the purpose of this study, only commuting activities and leisure time activities, i.e. walking, cycling, gardening, odd jobs and sports will be measured,...

Page 24. I deleted: Furthermore, a quality check will allow us to provide additional coaching to the counsellors if necessary.

Page 30: Figure 1 ihas been deleted and will be uploaded as a separate file.

Thank you for incorporating these changes in the final version of the manuscript.

Yours sincerely,
Iris Groeneveld